\begin{thebibliography}{10}

\bibitem{Abramsky87}
S.~Abramsky and C.~L.~(eds) Hankin.
\newblock {\em {Abstract Interpretation of Declarative Languages}}.
\newblock Ellis-Horwood, 1987.

\bibitem{Ariola89}
Z.~Ariola and Arvind.
\newblock {P-TAC: A Parallel Intermediate Language}.
\newblock In {\em Proceedings of the ACM Conference on Functional Programming
  Languages and Computer Architecture, London, UK}, pages 230--242. ACM,
  September 1989.

\bibitem{Ariola91}
Z.~Ariola and Arvind.
\newblock {Compilation of Id}.
\newblock In {\em Proc. of the Fourth Workshop on Languages and Compilers for
  Parallel Computing (LNCS 589)}. Springer-Verlag, August 1991.

\bibitem{Bloss88}
A.~Bloss.
\newblock {\em {Path Analysis and the Optimization of Non-strict Functional
  Languages}}.
\newblock PhD thesis, Dept. of Computer Science, Yale University, May 1989.

\bibitem{Bloss87}
A.~Bloss and P.~Hudak.
\newblock {Path Semantics}.
\newblock In {\em Mathematical Foundations of Programming Language Semantics
  (LNCS 298)}. Springer-Verlag, April 1987.

\bibitem{Burn86}
J.~Burn, C.~Hankin, and S.~Abramsky.
\newblock {Strictness Analysis for Higher-order Functions}.
\newblock {\em Science of Computer Programming}, 9, 1986.

\bibitem{Clack85}
C.~Clack and S.~L. Peyton-Jones.
\newblock {Strictness Analysis -- A Practical Approach}.
\newblock In {\em Proceedings of ACM Conference of Functional Programming
  Languages and Computer Architecture}. ACM, September 1985.

\bibitem{Coorg94}
S.~R. Coorg.
\newblock {Partitioning Non-strict Languages for Multi-threaded Code
  Generation}.
\newblock Masters Thesis, EECS, MIT, May 1994.

\bibitem{Culler93}
D.~E. Culler, S.~C. Goldstein, K.~E. Schauser, and T.~von Eicken.
\newblock {TAM --- A Compiler Controlled Threaded Abstract Machine.}
\newblock {\em {Journal of Parallel and Distributed Computing}},
  18(4):{347--370}, July 1993.

\bibitem{Garey79}
M.~R. Garey and D.~S. Johnson.
\newblock {\em {Computers and Intractability: A Guide to the Theory of
  NP-Completeness}}.
\newblock W.H. Freeman, San Francisco, CA, 1979.

\bibitem{Henderson80}
P.~Henderson.
\newblock {\em {Functional Programming: Applications and Implementation}}.
\newblock Prentice Hall, Englewood Cliffs, NJ, 1980.

\bibitem{Hoch91}
J.~E. Hoch, D.~M. Davenprot, V.~G. Grafe, and K.~M. Steele.
\newblock {Compile-time Partitioning of a Non-strict Language into Sequential
  Threads}.
\newblock In {\em Proc. Symp. on Parallel and Distributed Processing}, Dec
  1991.

\bibitem{Haskell90}
P.~Hudak and P.~Wadler~(editors).
\newblock {Report on the Programming Language {H}askell, A Non-strict Purely
  Functional Language ({V}ersion 1.0)}.
\newblock Technical Report YALEU/DCS/RR777, Yale University, Department of
  Computer Science, April 1990.

\bibitem{Hudak86}
P.~Hudak and J.~Young.
\newblock {Higher-order Strictness Analysis of the Untyped Lambda-Calculus}.
\newblock In {\em Proc. 12th ACM Symposium on Principles of Programming
  Languages}, pages 97--109. ACM, Jan 1986.

\bibitem{Ianucci88}
R.~A. Ianucci.
\newblock {\em {A Dataflow/von Neumann Hybrid Architecture}}.
\newblock PhD thesis, Dept. of EECS, MIT, May 1988.
\newblock (Also MIT LCS TR-418).

\bibitem{Mycroft80}
A.~Mycroft.
\newblock {The Theory and Practice of Transforming Call-by-need into
  Call-by-value}.
\newblock In {\em {International Symposium on Programming (LNCS 83)}}.
  Springer-Verlag, April 1980.

\bibitem{Nikhil90}
R.~S. Nikhil.
\newblock {Id Language Reference Manual Version 90.1}.
\newblock Technical Report CSG Memo 284-2, MIT Laboratory for Computer Science,
  545 Technology Square, Cambridge, MA 02139, July 15 1991.

\bibitem{Nikhil92}
R.S. Nikhil, G.M. Papadopoulos, and Arvind.
\newblock {*T: A Multithreaded Massively Parallel Architecture}.
\newblock In {\em Proc. 19th Annual Intl. Symp. on Computer Architecture}, May
  1992.

\bibitem{Schauser94b}
K.~E. Schauser.
\newblock {\em {Compiling Lenient Languages for Parallel Asynchronous
  Execution}}.
\newblock PhD thesis, Computer Science Div., Univ. of California at Berkeley.,
  May 1994.

\bibitem{Schauser94a}
K.~E. Schauser, D.~Culler, and S.~C. Goldstein.
\newblock {Separation Constraint Partitioning -- A New Algorithm for
  Partitioning Non-strict Programs into Sequential Threads}.
\newblock In {\em Proc. ACM Conference on Principles of Programming Languages
  (POPL)}, January 1995.

\bibitem{Schauser91}
K.~E. Schauser, D.~Culler, and von Eicken~T.
\newblock {Compiler-controlled Multithreading for Lenient Parallel Languages}.
\newblock In {\em Proc. Conf. on Functional Programming Languages and Computer
  Architecture}, August 1991.

\bibitem{Schmidt}
D.~A. Schmidt.
\newblock {\em {Denotational Semantics -- A Methodology for Language
  Development}}.
\newblock Allyn and Bacon, Boston, MA, 1986.

\bibitem{Traub86}
K.~R. Traub.
\newblock {A Compiler for the MIT Tagged-Token Dataflow Architecture}.
\newblock Technical Report TR-370, MIT Lab. for Computer Science, August 1986.
\newblock (MS Thesis, Dept. of EECS, MIT).

\bibitem{Traub89}
K.~R. Traub.
\newblock {Compilation as Partitioning: A New Approach to Compile Non-strict
  Functional Languages}.
\newblock In {\em Proc. of the ACM Conf. on Functional Programming and Computer
  Architecture}, pages 75--88. ACM, 1989.

\bibitem{Traub88}
K.~R. Traub.
\newblock {\em {Implementation of Non-strict Functional Programming
  Languages}}.
\newblock MIT Press, Cambridge, MA, 1991.

\bibitem{Traub92}
K.~R. Traub, D.~E. Culler, and K.~E. Schauser.
\newblock {Global Analysis for Partitioning Non-strict Programs into Sequential
  Threads}.
\newblock In {\em Proc. of the ACM Conf. on LISP and Functional Programming},
  June 1992.

\bibitem{Wadler87}
P.~Wadler.
\newblock {Strictness Analysis on Non-Flat Domains}.
\newblock In {\em \cite{Abramsky87}}.

\end{thebibliography}

\section{Introduction}
\label{section:introduction}
Functional programming languages can be divided into two classes:
{\em strict} and {\em non-strict}. 
In a non-strict language, functions may return values before all
their arguments are available, and data structures may be defined 
before all their components are defined. 
Many modern functional languages are non-strict; examples
include Haskell \cite{Haskell90} and Id \cite{Nikhil90}.
Such languages give greater
expressive power to the programmer than a strict language. 
Some of the languages provide non-strictness because of its cleaner
semantics. Others, like Id, also use non-strictness to generate
parallelism in a program.

Compiling non-strict languages to conventional microprocessors has been
the focus of many recent papers \cite{Traub92,Schauser91,Schauser94a}. 
Some features in the 
design of commercial microprocessors make it difficult to execute non-strict
programs directly. These processors are highly
tuned to execute a sequence of instructions efficiently by 
providing features like a large register set and deep pipelining. 
Unfortunately, this also implies
that there is a high penalty for dynamic scheduling of instructions.
As serial execution is interrupted,
registers have to be saved and restored, there
may be bubbles in the instruction pipeline etc. Synchronization
is also expensive in these processors since it 
has to be explicitly performed (in software) using instructions.

For programs in a non-strict language, it is not always possible 
at compile-time 
to specify an exact ordering of instructions. This is illustrated
in the following example written in Id.\\
Example 1:
\begin{verbatim}
def f a b = 
 {x = a + 1;
  y = b * 2
 in
   (x,y)};

{...                {...
 (c,d) = f 2 c       (c,d) = f d 1
...}                ...}
\end{verbatim}
The function {\tt f} takes two arguments {\tt a},{\tt b} and returns
two results {\tt x},{\tt y}. The set of statements enclosed by \{\ldots\}
form a recursive {\em let} block definition. There is no syntactic ordering
of the statements of the block. Also shown are calls to the function
{\tt f} in some other parts of the program (these are referred to
as {\em contexts} of {\tt f}). In both the contexts,
a result of {\tt f} is fed back into one of the function's arguments.

Consider the execution of the first context in a language
with strict semantics. The function {\tt f} is not called until
both of its arguments are available. In this case, it is not going
to be called until the variable {\tt c} gets a value. As
the value of {\tt c} depends on the call itself, this leads to a 
{\em deadlock} in the program; the variables {\tt c} and {\tt d}
never get defined.

In a non-strict execution, it is possible to call a function
even though not all of its arguments are available. Also, it is 
possible to return some results
even though not all of the function's results are available.
So, the function {\tt f} is called with the first argument
2 and the second argument $\bot$ (an {\em undefined} value). Then,
the value of {\tt x} can be computed to be 3. Now, the variable {\tt c}
gets defined to be the value 3, and the computation of {\tt y} and
{\tt d} can proceed. Hence, the non-strict execution produces
meaningful values for the variables {\tt c} and {\tt d}. Note that
in this execution, the $+$ instruction {\em must} have executed
before the $*$ instruction -- there is no other order that could
produce the correct results for {\tt f}.

Similarly, the second context produces a deadlock in a
strict execution, but produces valid results in a non-strict 
execution. However, this context requires that the $*$ instruction
execute before the $+$. Thus, depending upon the context, the 
instructions in {\tt f} can execute in either order. 
We have no choice but to compile these programs
 such that the two instructions execute independently,
when their data becomes available. Compiling {\tt f} in
this manner entails performing
dynamic scheduling/synchronization at each instruction -- leading
to an inefficient execution  on microprocessors.

Traub \cite{Traub88} 
made the significant observation that dependence analysis
could enable efficient compilation of some functions.
A simple example of such a function is given below.\\
Example 2:
\begin{verbatim}
def f a b =
 {x = a + b;
  y = a * b
 in
  (x,y)}
\end{verbatim}
In this function, we can infer that the dependence behavior
of both the $*$ and $+$ instruction is very similar: they 
both depend on the values of {\tt a} and {\tt b}. For this 
function, any context that feeds results back to arguments always
deadlocks, even in a non-strict execution. Thus, it is {\em safe}
to compile the function such that the two instruction execute
together, in some order.

Thus, the goal {\em partitioning} \cite{Traub88} is to infer
the possible dependencies in a program and use that information
to {\em sequentialize} parts of a program. This is achieved
by compiling each function into a set of {\em threads} while preserving
the function's non-strict semantics. Each
thread is a sequence of instructions and once a thread begins
execution, it does not need any dynamic scheduling/synchronization
during its execution. 
Threads of this form yield efficient implementation on conventional
architectures using an abstract machine such as TAM \cite{Culler93}.
They can also be implemented on parallel multi-threaded
hardware (\eg, *T \cite{Nikhil92}). 

Combining instructions into threads enables
more efficient use of registers and amortizes the overheard
of dynamic scheduling/synchronization over many instructions.
Partitioning attempts to make threads
as large as possible, to enable efficient execution of non-strict
programs on microprocessors. 


The strategy used to implement non-strictness 
(that is, functions
accepting {\em undefined} arguments and returning {\em partial} results) 
is called the {\em lenient} strategy \cite{Traub88}. 
This strategy gives flexibility 
to the programmer in using the recursive {\em let} block. It enables
circular dependencies to be resolved at run-time, instead of being 
statically resolved. Non-strictness in data-structures
also gives the programmer ability to create circular data-structures
functionally \cite{Traub88}.

Another 
strategy that provides non-strictness is the {\em lazy evaluation}
strategy, where the evaluation of an expression is delayed until
the execution cannot proceed without the value of that expression.
Unlike lenient evaluation, 
lazy evaluation also gives the user the ability to manipulate
infinite data structures \cite{Henderson80}. In this paper,
we are primarily interested in implementing a non-strict language
with the lenient strategy (though we do remark on the applicability
of the techniques developed here to implement lazy evaluation).

The contributions of this paper are:
\begin{itemize}
\item The definition of partitioning as a program transformation. 
\item Design of new basic block partitioning algorithms. Also, a
precise characterization of the complexity of the partitioning problem.
\item A new inter-procedural partitioning algorithm based on
abstract interpretation.
\item The proof of correctness of the partitioning
 algorithms using denotational semantics.
\end{itemize}

The rest of the paper is organized as follows. Section~\ref{section:syntax}
defines a simple non-strict kernel language and its semantics. Section~\ref{section:partitioning}
formally defines the partitioning problem and the notion of correctness 
of a partitioning. 
Section~\ref{section:basic} deals with the problem of partitioning 
basic blocks. 
Section~\ref{section:global} uses abstract interpretation techniques
to extend the basic block partitioning algorithm
to partition arbitrary blocks. 
Section~\ref{section:related} compares this paper with 
related work and section~\ref{section:conclusion}
concludes. In the appendix,
we present proof sketches of some of the theorems stated in this paper.



